Author’s response to reviews

Title: Simple Citrate anticoagulation protocol for Low Flux haemodialysis

Authors:

terina seow (123terina@gmail.com)
Eng Kuang Lim (lim.eng.kuang@alexandrahealth.com.sg)
Shun E Chen (chen.shune@alexandrahealth.com.sg)
Yang Gao (gg_yy_2001@yahoo.com.sg)
Min Er Liaw (liawminer@gmail.com)
Shimi Isaac (shimi_issac@yahoo.com)

Version: 4 Date: 09 Nov 2017

Author’s response to reviews:

1. Trial Registration

We believe that your study requires trial registration and we would like to ask for the trial registration number and date of registration to be included at the end of the abstract. Have registered trial retrospectively.

reply: Trial registration number and date now added to end of abstract

2. In a previous revision we asked that all cases of 'mls' be revised. Thank you for addressing this in the text. We note, however, that mls still appears in the figures. Please revise.

reply: Changed in Figure 1
3. Title page

Please ensure a title page appears at the start of your manuscript. The title page should include the full names, institutional addresses and email addresses for all authors.

reply: Have put title page at start of manuscript